Author's response to reviews

Title: Using a state cancer registry to recruit young breast cancer survivors and high-risk relatives: a protocol of a randomized trial testing the efficacy of a targeted versus a tailored intervention to increase breast cancer screening

Authors:

Maria C Katapodi (mkatapo@umich.edu)
Laurel L Northouse (lnortho@umich.edu)
Ann M Schafenacker (annschaf@umich.edu)
Debra Duquette (duquetted@michigan.gov)
Sonia A Duffy (bump@umich.edu)
David L Ronis (dronis@umich.edu)
Beth Anderson (andersonb@michigan.gov)
Nancy K Janz (nkjanz@umich.edu)
Jennifer McLosky (mcloskyj@michigan.gov)
Kara J Milliron (kmilliro@umich.edu)
Sofia D Merajver (smerajve@umich.edu)
Linh M. Duong (gqi9@cdc.gov)
Glenn Copeland (copelandg@michigan.gov)

Version: 3 Date: 14 December 2012

Author's response to reviews: see over